Author's response to reviews

Title: Influenza vaccination coverage rates among adults before and after the 2009 influenza pandemic and the reasons for non-vaccination in Beijing, China: a cross-sectional study

Authors:

Shuangsheng Wu (wushuangsheng0606@126.com)
Peng Yang (clairecaoyang@163.com)
Haiyue Li (0418012058@163.com)
Chunna Ma (ma_chun_na@126.com)
Yi Zhang (zps347@163.com)
Quanyi Wang (bjcxcxm@126.com)

Version: 10 Date: 12 June 2013

Author's response to reviews: see over
Dear Editors:

My manuscript, ‘Influenza vaccination coverage rates among adults before and after the 2009 influenza pandemic and the reasons for non-vaccination in Beijing, China: a cross-sectional study’, was revised according to your comments, and was edited by a professional language editing service.

Many thanks for your suggestion. Thanks very much again for your attention to our paper.

Sincerely yours,

Shuangsheng Wu